MANUSCRIPT NUMBER 29114
Dear Editor,
please find enclosed our revised manuscript 29114 “Gastric and duodenal polyps in FAP patients: conventional endoscopy vs virtual chromoendoscopy (FICE) in dysplasia evaluation” that we have been invited to write for the “World Journal of Clinical Oncology”.
The manuscript has been reviewed according to reviewers request as explained in the point by point reply letter.

I truly hope that you will find this paper suitable for publication.

Waiting for your answer,
Sincerely,

Dr. Simone Polvani

Dr. Simone Polvani, PhD
Dept. of Experimental and Clinical Biomedical Sciences “Mario Serio”, University of Florence,viale Pieraccini 6, 50139 Florence, Italy,
email: simone.polvani@unifi.it
tel. 0039 055 2758 115
Reply to reviewer 1 (503623):
We thank the reviewer for the appreciation of our work.

Reply to reviewer 2 (58401):
We thank the reviewer for the useful comments. According to reviewer request we have added two new references (for vitamin B12 and FAP and to a recent paper published on Lancet regarding the FICE).

Name: Simone Polvani
Signature: [Signature]
Date: 06/01/16